Author's response to reviews

Title: The Tinnitus Research Initiative (TRI) database: A new approach for delineation of tinnitus subtypes and generation of predictors for treatment outcome

Authors:

Michael Landgrebe (michael.landgrebe@medbo.de)
Florian Zeman (Florian.Zeman@klinik.uni-regensburg.de)
Michael Koller (Michael.Koller@klinik.uni-regensburg.de)
Yvonne Eberl (Yvonne.Eberl@klinik.uni-regensburg.de)
Markus Mohr (mm@manathea.de)
Jean Reiter (jean.reiter@medbo.de)
Susanne Staudinger (susanne.staudinger@medbo.de)
Goeran Hajak (goeran.hajak@medbo.de)
Berthold Langguth (berthold.langguth@medbo.de)

Version: 2 Date: 17 May 2010

Author's response to reviews:

Dear Mrs Pafitis, dear Ladies and Gentlemen,

according to your email from May 15th, we have modified our above mentioned manuscript, so that it now follows the instructions for a type 3 correspondence. We also laid more emphasis on the process to develop the database and described in more detail, which kind of problems we faced and how they were solved. Since the TRI database is unique in tinnitus research by its multicenter, multinational approach, we do believe, that it will be of great interest for the readership of your journal!

Sincerely yours,

Michael Landgrebe, MD